Author's response to reviews

Title: Multilevel analysis of systolic blood pressure and ACE gene I/D polymorphism in 438 Swedish families - a public health perspective

Authors:

Juan Merlo (juan.merlo@med.lu.se)
Kristina Bengtsson-Bostrom (kristina.bengtsson@endo.mas.lu.se)
Ulf Lindblad (Ulf.Lindblad@med.lu.se)
Lennart Rastam (lennart.rastam@med.lu.se)
Olle Melander (olle.melander@med.lu.se)

Version: 2 Date: 3 November 2005

Author's response to reviews:

According the Editorial the manuscript has been transferred to BMC Medical Genetics, Here is a revised version containing formatting changes suitable for this Journal